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• All chambers included, but ATLAS


• No major problems during the shif

Stability run 234



Stability run 234



HV scan 1003

Wednesday 13th, November

Trip: current > I0set (50 uA)



HV scan 1004

Wednesday 13th, November

I0set 70 uA during HV scan



HV scan 1002 from last week

Wednesday 6th, November

• No huge differences observed w.r.t. this week HV scans



Instructions for the next shifter

Detectors to be included: ALICE,	 EPDT, 	RE1-1,	 SHiP, KODEL-H 

For current	 scan 	use template 10	04

For stabili	ty sca	n use template 	234


RE1_1 I0 set to 50 uA by default -> set 70 uA during HV scan


